Author’s response to reviews

Title: Clinical Outcome of Endovascular Coil Embolization For Cerebral Aneurysms In Asian Population In Relation To Risk Factors: A 3-Year Retrospective Analysis

Authors:
Saima Ahmad (masterinfluencer@gmail.com)

Version: 1 Date: 09 Nov 2019

Author’s response to reviews:

1. Gender was removed from the main table.
2. Corrected.
3. Author name fully written.